Author's response to reviews

Title: Vascular basis of mucosal color

Authors:

Johannes Kleinheinz (joklein@uni-muenster.de)
Andre Buchter (Andre.Buechter@ukmuenster.de)
Thomas Fillies (Thomas.Fillies@ukmuenster.de)
Ulrich Joos (joos@uni-muenster.de)

Version: 4 Date: 24 June 2005

Author's response to reviews:

Table 2 and Table 3 were changed into Figure 10 and Figure 11. Tables were removed from the manuscript and legends have been added.